Author's response to reviews

Title: Association of a probiotic to a Helicobacter pylori eradication regimen does not increase efficacy or decreases the adverse effects of the treatment: a prospective, randomized, double-blind, placebo-controlled study.

Authors:

Tomás Navarro-Rodrigues (tomasnavarro@uol.com.br)
Ricardo C Barbuti (rbarbuti@terra.com.br)
Rejane Mattar (rmattar@hcnet.usp.br)
Joaquim P Moraes Filho (joaquim.prado@uol.com.br)
Maricê N Oliveira (monolive@usp.br)
Cristina SB Bogsan (cris.bogsan@uol.com.br)
Décio Chinzon (clinica.chinzon@uol.com.br)
Jaime N Eisig (eisig@uol.com.br)
Fernando M Silva (fernando.marcuz@hc.fm.usp.br)

Version: 4 Date: 8 March 2013

Author's response to reviews:

Dear editor

According your suggestion, the paper was submitted for editing with Edanz. We believe that it is now suitable for publication. Thank you for your attention,

The authors.